Author’s response to reviews

Title: Comparison of procalcitonin and high-sensitivity C-reactive protein for the diagnosis of sepsis and septic shock in the oldest old patients

Authors:

Hongmin Zhang (ozohom@163.com)
Xiaoting Wang (SYZZYX2015@sina.com)
Qing Zhang (zhqmt@163.com)
Ying Xia (lnxlyh@126.com)
Dawei Liu (dwliu2015@sina.com)

Version: 3 Date: 19 Jul 2017

Author’s response to reviews:

Not applicable